Author’s response to reviews

Title: National Essential Medicines List and Policy Practice: A Case study of China’s Health Care Reform

Authors:

Xin Tian (tianxin_66610@sina.com)
Yaran Song (shakira_sung@yahoo.cn)
Xinping Zhang (xpzhang602@163.com)

Version: 2 Date: 16 April 2012

Author’s response to reviews:

The date used in manuscript was openly available in the internet, and the study did not involve any ethical problems.